Author’s response to reviews

Title: Is informed consent related to success in exercise and diet intervention as evaluated at 12 months? DR’s EXTRA study

Authors:

Helena Länsimies-Antikainen (helena.lansimies-antikainen@uef.fi)
Anna-Maija Pietilä (anna-maija.pietila@uef.fi)
Tomi Laitinen (tomi.laitinen@kuh.fi)
Vesa Kiviniemi (vesa.kiviniemi@uef.fi)
Rainer Rauramaa (rainer.rauramaa@uef.fi)

Version: 5 Date: 6 June 2010

Author’s response to reviews:

UNIVERSITY OF EASTERN FINLAND
6 June 2010

Scientific Editor Rachel Neilan,
BMC-series Journals
BioMed Central

Dear Ms Neilan,

Thank you once again for the acceptance of our manuscript entitled “Is informed consent related to success in exercise and diet intervention as evaluated at 12 months? DR’s EXTRA study” (Ms. 1704541388281585).

We have taken into account reviewer’s kind comment and made the change in manuscript (page 10, line 1). In addition, we have taken into account the Editorial Production Team suggestions:

1. We have removed all figure and placement text (page 5, last line; page 7, paragraph 2, last line; page 9, paragraph 2, last line; page 10, paragraph 2 and 3, last lines). The figures and tables are cited in the manuscript (new text at page 10, paragraph 2 and 3, last lines).

2. The table titles are now one sentence long. All other text is placed below the table (see tables 2 and 3).

We have checked our manuscript as a whole. We are very grateful if you can do the final layout of our figures.

Thank you for your kind collaboration and have a nice summer.

Yours sincerely,

Helena Länsimies-Antikainen